Author's response to reviews

Title: The effectiveness of Tai Chi on physical and psychological well-being of college students: study protocol for a randomized controlled trial

Authors:

Guohua zheng (zhgh_1969@aliyun.com)
Xiulu Lan (fcrystal_baby@163.com)
Moyi Li (limovi123@aliyun.com)
Kun Ling (Lingkun6101@163.com)
Hui Lin (376775611@qq.com)
Lidian Chen (lidianchen87@163.com)
Jing Tao (taojing01@163.com)
Junzhe Li (784034587@qq.com)
Xin Zheng (zhengxin123366816@126.com)
Bai Chen (327367260@qq.com)
Qianying Fang (117806500@qq.com)

Version: 2
Date: 6 November 2013

Author's response to reviews: see over
Trials
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear editor:

We are submitting our study protocol, *The effectiveness of Tai Chi on physical and psychological well-being of college students: study protocol for a randomized controlled trial*, for consideration for publication in the trial.

We should like to inform you of the following:

- This work has not been published or submitted for publication elsewhere.
- All authors are in agreement with the content and publication of the manuscript.
- Each author declares that he/she did not receive financial support for this research, and does not have relationships that pose a conflict of interest.
- The authors’ respective institutions have approved of the submission of this manuscript.

Thank you for considering our manuscript. We look forward to hearing from you.

Sincerely,
Xiulu Lan